Author's response to reviews

Title: Reasons for non-participation in a parental program concerning underage drinking: a mixed-method study

Authors:

Camilla Pettersson (camilla.pettersson@oru.se)
Margareta Lindén-Boström (margareta.linden-bostrom@orebroll.se)
Charli Eriksson (charli.eriksson@oru.se)

Version: 3 Date: 14 December 2009

Author's response to reviews: see over
Dear Editor

We have made the formatting changes that the editorial team suggested:

- We have re-layout the table 5 and 6 and deleted vertically text
- The information from Table 6 is now showed in two tables instead of one.
- We have changed all underlining text.
- We have removed all visible vertical lines from the tables.
- We have changed the typeface in the figure to Arial 12pt.

We look forward to hearing from you again

Best regards
Camilla Pettersson